Author's response to reviews

Title: The development and evaluation of the Australian Child and Adolescent
Recommended Food Score: a cross-sectional study

Authors:

Skye Huxley (Skye.Huxley@uon.edu.au)
Jane F Watson (Jane.Watson@newcastle.edu.au)
Tracy L Burrows (Tracy.Burrows@newcastle.edu.au)
Maya Guest (Maya.Guest@newcastle.edu.au)
Clare E Collins (clare.collins@newcastle.edu.au)

Version: 4 Date: 16 July 2012

Author's response to reviews:

This was submitted in the 14th July